Author’s response to reviews

Title: Effect of Fagonia Arabica (Dhamasa) on in vitro thrombolysis

Authors:

Sweta Prasad (sweta_ciims@yahoo.com)
Rajpal S Kashyap (raj_ciims@rediffmail.com)
Jayant Y Deopujari (jy_deopujari@rediffmail.com)
Hemant J Purohit (hemantdrd@rediffmail.com)
Girdhar M Taori (taorigm_ciims@yahoo.co.in)
Hatim F Daginawala (hfd_ciims@rediffmail.com)

Version: 4 Date: 26 September 2007

Author’s response to reviews:

To,
Elizabeth C Moylan PhD
Senior Editor
BioMed Central
Middlesex House
32-42 Cleveland Street
London
W1T 4LB
UK

Dear Madam,

Please find enclosed copy of the revised manuscript entitled Effect of Fagonia Arabica (Dhamasa) on in vitro thrombolysis (MS: 1588256974135811) along with the reply to the reviewer’s comments. We are highly thankful to both the reviewers for their highly critical review of our article.

With best regards,

Dr. H.F. Daginawala

Note: Acknowledgement included in the manuscript

Reviewer 2
Thomas Stief

Minor compulsory revision

Comment: The fibrinolysis results should be given only with one position behind
the comma, the assay system used is not as precise as to use two positions.

Reply: The fibrinolysis results are now represented upto one decimal point after rounding off. The corrections are made wherever the percentage of clot lysis is mentioned viz., abstract, results, discussion, table 1, figure 1 and the respective legends.